Author's response to reviews

Title: Hardiness Personality and Burnout Syndrome among nursing students of three Brazilian universities - An analytical study

Authors:

Rodrigo M Silva (marques-sm@hotmail.com)
Carolina T Goulart (carolinatonini@yahoo.com.br)
Luis Felipe D Lopes (lflopes67@yahoo.com.br)
Patrícia M Serrano (paserrano@usp.br)
Ana Lúcia S Costa (ananascosta@usp.br)
Laura A Guido (lguido344@gmail.com)

Version: 5 Date: 20 January 2014

Author's response to reviews: see over
COVER LETTER

Editor of *BMC Nursing*.

We revised the document named “Hardiness Personality and Burnout Syndrome among nursing students of three Brazilian universities – An analytical study” according to the suggestions provided by referees, as follows:

Referee 1

The English writing review was made by a native professional from USA.

Referee 2

The English writing review was made by a native professional from USA.

The importance of Burnout Syndrome and Hardiness Personality for the nursing field was described in introduction, discussion and conclusion of the paper.

We explained to the readers each result presented in results of the paper.

__________________________
Rodrigo Marques da Silva

__________________________
Carolina Tonini Goulart

__________________________
Luis Felipe Dias Lopes

__________________________
Patrícia Maria Serrano

__________________________
Ana Lúcia Siqueira Costa

__________________________
Laura de Azevedo Guido